\begin{table}%t1
\caption{\label{tab1}Observation log and properties of observed objects.
$V_0$ is from Evstigneeva et~al. (\cite{evst06}) and $(V-I)_0$ from Mieske et~al. (\cite{mies06}) (except $(V-I)_0$ for FCC303 which is from Karick et~al.
\cite{kari03}). Both, $V_0$ and $(V-I)_0$ are corrected for foreground dust
extinction (Schlegel et~al.  \cite{schl98}). The [Fe/H] values are from Mieske
et~al. (\cite{mies06}).}
\small%\centerline
{
\begin{tabular}{p{0.9cm}p{2.5cm}ccp{0.5cm}p{0.9cm}@{\hspace{2mm}}p{0.6cm}lccp{0.6cm}p{0.8cm}}\\
\hline\hline
\noalign{\smallskip}
Name & FCSS$^a$ name & $\alpha$(2000) & $\delta$(2000) & $V_0$ & $(V-I)_0$ &
 [Fe/H] & Date & Exp.~time & Seeing & {\it S/N}$^b$ & {\it S/N}$^b$ \\
 & & [h:m:s] & [$^\circ$:$\arcmin$:$\arcsec$] & [mag] & [mag] & [dex] & & [s] &
[$\arcsec$] & 4900~$\AA$ & 5900~$\AA$ \\
\hline
 \noalign{\smallskip}
FCC303$^c$ &              & 3:45:14.10 & $-$36:56:12.0 & 15.90 & 1.09 & &
2000~Dec.~20 & $3\times1004$ & 0.80 & 3.6 & 4.2 \\
UCD2 & J033806.3$-$352858 & 3:38:06.33 & $-$35:28:58.8 & 19.12 & 1.14 & $-$0.90
 & 2000~Dec.~23 & $5\times1800$ & 0.55 & 4.5 & 5.6 \\
UCD3 & J033854.1$-$353333 & 3:38:54.10 & $-$35:33:33.6 & 17.82 & 1.22 & $-$0.52
 & 2000~Dec.~20 & $3\times1000$ & 0.80 & 4.7 & 5.9 \\
UCD4 & J033935.9$-$352824 & 3:39:35.95 & $-$35:28:24.5 & 18.94 & 1.13 & $-$0.85
 & 2000~Dec.~24 & $5\times1800$ & 0.60 & 4.8 & 5.6 \\
UCD5 & J033952.5$-$350424 & 3:39:52.58 & $-$35:04:24.1 & 19.40 & 1.00 & &
2001~Feb.~12 & $2\times3285$ & 0.60 & 4.4 & 5.2 \\
UCD5 & & & & & & & 2001~Feb.~13 & $2\times3285$ & 0.55 & 4.8 & 5.8 \\
UCD5 & & & & & & & 2001~Aug.~19 & $1\times3285$ & 0.55 & 4.6 & 5.1 \\
UCD5 & & & & & & & 2001~Sep.~20 & $1\times3119$ & 0.80 & 5.3 & 6.3 \\
UCD5 & & & & & & & 2001~Nov.~19 & $1\times3285$ & 0.85 & 5.2 & 6.4 \\
\hline
\end{tabular}}
\vskip0.1cm
$^a$Fornax Cluster Spectroscopic Survey (Drinkwater et al. \cite{drin99});
$^b$measured for a single spectrum per resolution element ($= 0.03$~\AA);
$^c$Fornax Cluster Catalog (Ferguson \cite{ferg89a}).
\end{table}